Author's response to reviews

Title: Determination of the human spine curve based on laser triangulation

Authors:

Primoz Poredos (primoz.poredos@fs.uni-lj.si)
Dusan Celan (dusan.celan@ukc-mb.si)
Janez Mozina (janez.mozina@fs.uni-lj.si)
Matija Jezersek (matija.jezersek@fs.uni-lj.si)

Version: 4 Date: 6 January 2015

Author's response to reviews: see over
Dear Krisha Mae Natan,

thank you very much for sending us the second revision of our manuscript »Determination of the human spine curve based on laser triangulation« (MS: 1674272075139290). We are glad that both reviewers now agree with publication of the manuscript.

Regarding the Editors Comments please find below our point-by-point responses:

**Editorial Comment 1:**
“Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.”

**Our response:**
We changed the fifth paragraph in section Methods: “All patients consented to participate in the presented research after obtaining an informed written consent. In case of children participation in the research, the informed written consent was obtained either by their parent or guardian.”

**Editorial Comment 2:**
“Consent to publish - We note that figures 2, 3, 4 and 7 contain images of patients involved in the study. Please clarify whether you have obtained consent to publish these images from the participants involved and include a statement to this effect at the end of your paper.”

**Our response:**
At the end of our paper we added section Consent with the following contents: “Written informed consent was obtained from patients for publication of this research article including accompanying images.”

With kindest regards,

Primož Poredoš
Dušan Čelan
Janez Možina
Matija Jezeršek

Ljubljana, January 6th, 2015